Author’s response to reviews

Title: Pharmacological potential of Bidens pilosa L. and determination of bioactive compounds using UHPLC-QqQ-MS/MS and GC/MS

Authors:

Garima Singh (garima.singh106@gmail.com)
Ajit Passari (ajit.passari22@gmail.com)
Vincent Leo (vincentvineethleo@gmail.com)
Sarathbabu Subbarayan (sarth07bio@gmail.com)
Pratibha Singh (garimasingh@mzuonline.in)
Brijesh Kumar (gbrikum@yahoo.com)
Bhim Singh (bhimpratap@gmail.com)
Hauzel lalhlenmawia (hlenmawia@gmail.com)
Nachimuthu Senthil kumar (nsmzu@gmail.com)

Version: 2 Date: 02 Nov 2017

Author’s response to reviews:

Authors responses to reviewer(s) comments

Authors are very much thankful to all the reviewers and editor for their valuable comments. We have revised manuscript as per your suggestions and incorporated the correction in the revised manuscript. Please find below the answers and rebuttals of the queries raised by editor/reviewers.

Comment 1

Please include the email addresses of all authors on the title page of your manuscript.

Author response

Email addresses of all authors have been included on the title page.

Comment 2
Please revise your Authors' contribution section to confirm all authors have read and approved of the final manuscript.

Author response

Authors’ contribution section has been revised.

Comment 3

Please add a Funding section to your Declarations.

Author response

Funding section has been included in declarations